DEP AR TMENT OF ECONOMICS

Econ 301

Interme diate Micr o e c onomic The ory

1 Outline:

Instructor: Kevin Wainwright 3681 WMX

Phone: 291-4438

Office Hours: 7:30pm-8:30pm T ues. and Thurs.

E-mail: wainwrig@sfu.c a

2 T ext b o oks:

Micro economics (3rd Ed.) by Eaton and Eaton (and

Study Guide)

3 T opics :

3.1 In tro duction and math review

Eaton: chapter 1; Mathematic al Handout

3.2 Consumer theory

1. Utilit y Theory: Eaton: chapter 2

2. Indifference curv e analysis: Eaton chapter3

3. Demand curv es, price consumption curv es, elastic-

it y: Eaton chapter 3

4. Applications and extensions:

(a) T axation, t w o-part pricing: Eaton chapter 4

(b) Consumer surplus, CV and EV: Eaton chapter

4; Handout

(c) Lab our Supply , rationing: Eaton chapter 4 and

chapter 14 (se ction 14.4 only)

3.3 Pro duction and Costs

1. Short run pro duction and costs:

Eaton chapter 7

(a) The pro duction function to the v ariable cost

function

(b) Pre-pro duction costs

(c) Opp ortunit y cost

2. Long run pro duction and costs:

Eaton chapter 8

(a) The cost minimization problem

(b) The marginal rate of tec hnical substitution

(c) F actor demand functions

3.4 P erfect Comp etition

Eaton chapter 9

1. Profit maximization and industry supply

2. Short run equilibrium

3. En try , exit, and long run equilibrium

4. Comparativ e statics: c hanges in v ariable costs, fixed

costs, taxes

5. Increasing and decreasing cost industries

3.5 Monop oly

Eaton chapter 10

3.6 Oligop oly and Game Theory

Eaton chapter 11

3.7 Other T opics (one or more of the

f ollo wing):

1. Exp ected utilit y: Eaton chapter 5

2. Externalities, Public go o ds and Natural Resources:

Eaton chapters 15 and 17

3. Pro duct Differen tiation: Eaton chapter 12